Author’s response to reviews

Title: Abortion in Nepal: Perspectives of a cross-section of sexual and reproductive health and rights professionals

Authors:

Claire Rogers (hello@clairerogers.com.au; claire.rogers@postgrad.curtin.edu.au)

Sabitri Sapkota (Sabitri.Sapkota@mariestopes.org)

Anita Tako (aneetako@gmail.com)

Jaya Dantas (jaya.dantas@curtin.edu.au)

Version: 4 Date: 10 Feb 2019

Author’s response to reviews:

Amendments made as requested.

1. Please specify in the Methods and Ethics approval and consent to participate sections whether you obtained written or verbal consent from the participants in your study. If verbal, please state the reason and whether the ethics committee approved this procedure.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.